Author's response to reviews

Title: A contingent valuation study to estimate the parental willingness-to-pay for childhood diarrhoea and gender bias among rural households in India

Authors:

Dr Mo A Amin (moa@ccohta.ca)
Farhana Khondoker (farhana.khondoker@sympatico.ca)

Version: 4 Date: 2 Jun 2004

Dear Editor,

We have made all formatting changes as suggested by Valery Abramovv and Anna Howarth. We are happy with all these changes. Please find attached the final corrected version of the manuscript for your kind consideration.

Many thanks,

Mo Amin.